Author’s response to reviews

Title: EXPLORING THE RELATIONSHIP BETWEEN THE BIG FIVE PERSONALITY CHARACTERISTICS AND DIETARY HABITS AMONG STUDENTS IN A GHANAIAN UNIVERSITY.

Authors

Freda Intiful (fdintiful@chs.edu.gh)
Emefa Oddam (emefatoddam@yahoo.com)
Irene Kretchy (ikretchy@ug.edu.gh)
Joana Quampah (quampah73@yahoo.com)

Version: 1 Date: 09 Jan 2018

Author’s response to reviews:

1. The findings of previous works have been included in the introduction as suggested.
2. The procedure for obtaining the sample size has been outlined under the heading ‘study participants and sample size’. This should explain how the sample size was arrived at.
3. Cronbach Alpha coefficient have been calculated for all the questionnaires that were used and reported at the results section.
4. Grammar and punctuations have been revised.
5. The references have been corrected for errors. The query on out of date citations could not be adequately addressed because not many current articles in this area were found to be relevant to the topic under discussion.